RESEARCH

Open Access

Understanding sexual and reproductive health needs of young women living in Zika affected regions: a qualitative study in northeastern Brazil



Debora Diniz¹, Moazzam Ali^{2*}, Ilana Ambrogi³ and Luciana Brito^{4*}

Abstract

Background: In 2016, the World Health Organization declared a Public Health Emergency of International Concern due to Zika's association with microcephaly and other neurological disorders. Brazil was the epicenter of this epidemic and the most affected region has the lowest Human Development Index and the highest rates of adolescent pregnancy. Despite the end of the epidemic, Brazil continues to be the epicenter of Zika illness. This study examined the barriers faced by young women who seek sexual and reproductive health (SRH) care services living in affected areas and their attitudes towards SRH needs and the available services.

Methods: Individual semi-structured interviews were conducted with 22 young women, aged 14–24 years in three Zika affected municipalities in the Brazilian Northeast. This qualitative research used thematic analysis for data analysis.

Results: Almost half (n = 10) of the participants had their first pregnancy during adolescence (from 12 to 19), all of which were unintended. Lack of information and barriers to access family planning were found to contribute to the unmet need for contraception. Participants reported knowledge gaps about contraception. Zika was not considered a health concern and participants were unaware of the possibility of Zika's sexual transmission.

Conclusions: The young women's knowledge and attitudes towards their SRH needs highlight the barriers to access care. It also implies that comprehensive, biopsychosocial and political, understanding is necessary in order to adequately provide SRH to this population and meet their needs. The government should place women at the center of any public health response to an emergency affecting women of reproductive age and focus on improving access to information and family planning services in a culturally and age appropriate manner.

Keywords: Zika virus, Sexual and reproductive health, Public health, Young women, Brazil

Plain English summary

Congenital Zika Syndrome is a novel syndrome due to in-utero Zika virus infection. It was discovered after Brazil had become the epicenter of the Zika epidemic in 2016. The mosquito born Zika virus has been shown to cause microcephaly and other severe abnormalities in the developing fetus and child. It can also affect the

* Correspondence: alimoa@who.int; l.brito@anis.org.br

²Department of Reproductive Health and Research (RHR), World Health Organization, Avenue Appia 20, CH-1211 Geneva 27, Switzerland ⁴Director of Research at Anis - Institute of Bioethics, Caixa Postal 8011, Brasília, DF 70094-971, Brazil

Full list of author information is available at the end of the article

pregnancy. Due to its severe effects, the World health Organization declared a Public Health Emergency of International Concern in 2016 and efforts were made to learn more about the consequences of this infection. It has been found that the Zika virus is also transmitted sexually and that the most affected are young, poor, underserved afro and native Brazilian women.

This study investigates the sexual and reproductive health (SRH) needs of young women living in Zika affected areas. Women interviewed were seeking SRH services at three different northeastern cities' public health services between November 2017 and July 2018. Ten out



© The Author(s). 2020 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated. of 22 had their first pregnancy while still in adolescence; all of these pregnancies were unintended. Zika was not a health concern and they were not aware that Zika could be sexually transmitted. Lack of information and barriers to access aggravated their unmet SRH needs.

In order to improve SRH of women at most risk, they need to be placed at the center of any public health response to an emergency. It is only by understanding their needs and the obstacles they face when seeking care that these issues will be adequately addressed.

Background

Zika virus is a mosquito borne virus that has been shown to cause microcephaly and other severe brain anomalies in newborns when the mother is infected prenatally [1, 2]. The Northeast of Brazil was the epicenter of the Zika epidemic; and for the fifth time in history the World Health Organization (WHO) declared, on February 1st, 2016, a Public Health Emergency of International Concern (PHEIC) due to the association of Zika with microcephaly and other neurological disorders. On November 18th of that year, the PHEIC was lifted [3]. The Brazilian Ministry of Health also encouraged women, planning to postpone pregnancies, to seek contraceptive methods at Basic Health Units [4]. This underserved Northeast Brazilian geographic region has the highest adolescent pregnancy rates, lowest levels of formal schooling and scarce access to health services [5, 6]. This region has several social-economic challenges, where most of the population does not have access to basic services such as running water, access to health and information and where the mosquito vector Aedes *aegypti* and tropical diseases like dengue have been part of everyday life for generations [6]. All these factors influence the health of the population and make it vulnerable to outbreaks and other health related challenges.

Given the consequences of Zika infection on pregnancy and fetal development, it has been understood as an enduring public health task. However, most of the efforts in response to the epidemic were related to vector control, development of adequate methods for diagnosis and the possibility of a vaccine. Little attention was given to the social impacts of Zika virus and how to provide information to a population with low literacy and little access to health services [7].

There is a general lack of information about sexual and reproductive health (SRH) and there are barriers to access public health facilities for family planning. The Brazilian Public Health System (SUS) offers several contraceptive methods for free [8]. Although studies show that Brazil has a relatively high modern contraceptive prevalence, there is low use of long-acting reversible contraceptives (LARCs) in the public sector, and inequities exists [9, 10]. In 2010, around 50–55% of all births in Brazil resulted from unintended pregnancies [11]. It is estimated that in Brazil, one in five children are born to adolescents between the ages of 10 and 19 years old [12–14]. Despite the observed decline in live births in Brazil since the emergence of the epidemic, there were no changes in the health policies and no evidence that there was an increase in demand for long-term contraceptive methods [12, 14]. These findings demand attention and evaluation in order to develop evidence-based guidance for strengthening health systems and to prevent harmful practices.

The Zika outbreak has exposed how access to sexual and reproductive health and its needs are a major public health issue, but there is a dearth of social and qualitative research. This paper aims to describe the young women's perceptions and their needs for sexual and reproductive health services as well as the barriers they face when seeking sexual health and reproductive planning in Brazilian public health care facilities located in regions most affected by the Zika virus. Although our study might not represent the entire population's view on the matter, it highlights the voices of the women most at risk of the negative impacts of the outbreak.

Methods

Design

This qualitative research used the thematic analysis method. Our aim was to investigate in a purposive sample and through in-depth analysis a conceptual description of young women's sexual and reproductive health needs as well as the barriers they faced trying to access health care services. Because of that, this study did not intend to specify frequencies, nor produce generalizing statistical inferences.

Recruitment of participants

The participants of this qualitative study were young women seeking sexual and reproductive health services in the Brazilian public health facilities of three Zika affected municipalities in the Northeast of Brazil: São Luís and Balsas in Maranhão state, and Santana do Ipanema in Alagoas state. Participants' ages ranged between 14 and 24 years old. This study will refer to its the participants which includes adolescents (aged 10-19) and youths (aged 15-24), as young women. The study setting was selected based on the three Brazilian health care facilities place-based Sentinel Sites Evaluation (SSE) activities established by the research study sites WHO Human Reproduction Program (HRP). Health care workers invited women seeking sexual and reproductive health services at these sites between November 2017 and July 2018 to participate in the study. An informed opt-in procedure was used, participants were only contacted by the research team after providing their consent for the contact to the health care workers. Data was stratified by age and only women younger than 24 years old were considered for this analysis.

Procedure and data analysis

Women researchers performed individual, in-depth interviews with 22 participants. The interviews lasted from 40 to 90 min. They were semi-structured interviews and followed a topic guide, but it took place as a conversation in which the researchers promoted a safe, comfortable environment to enable a comprehensive and candid record. Initial questions were on sociodemographic data, such as age, color, ethnicity, occupation, level of education, marital status, number of children, religion if any, and address. Questions about their attitude and knowledge on Zika virus, CZS, sexual transmission of Zika, contraception, abortion, pregnancy, and SRH needs and services, followed. The participants were invited to choose the places of the interviews, most of them preferred their homes (n = 20).

After each interview was transcribed, microanalysis was used as the analysis strategy. Microanalysis is a type of detailed analysis of data in which line by line or sentence by sentence of documents are examined at the beginning of the study [15]. With an inductive approach to the analysis initial codes were generated. Codes were collated and those with similar pattern of meaning were grouped in order to identify themes [15, 16]. The coding process was used in order to focus the data analysis on what emerged from the empirical evidence. The coding process was systematically discussed between the researchers and was initiated after each interview was transcribed in order to identify core information during the data collection phase. Transcribed interviews collected by two researchers were subsequently coded accordingly [15, 16]. Coded interviews and the data were tabulated in order to identify the patterns and review themes. Standards of coding were compared, and, in case of discrepancy, the coding method was discussed by the research team. In this analysis two themes emerged: lack of information and barriers of access to reproductive and sexual health services.

This study protocol was approved by the University of Brasilia Research Ethics Committee for Humanities and Social Sciences – CAAE: 66502017.0.0000.5540 and WHO Ethics Review Committee. All the participants provided informed consent in order to participate in the study in accordance with Brazilian regulations which guarantee autonomy and confidentiality about the sexual and reproductive health decisions and practices of the individual.

Results and discussion

Participant characteristics

Participants in the three municipalities had similar characteristics. The majority of them declared themselves as Afro-Brazilians (n = 18). Of the participants who declared having a religion (n = 17), Catholicism (n = 10)and Evangelicalism (n = 7) were reported. Only one participant said she had no religion. Eight participants had a very low level of formal schooling: they stopped their studies while in elementary school, a higher proportion than expected for this age group. According to the Brazilian National Household Survey, PNAD 2015, the national rate of adolescents between 17 and 19 years old with incomplete elementary education was less than 15% [6]. Among participants with very low formal schooling, only one had not experienced an adolescent pregnancy. Twelve participants completed high school or discontinued their studies during high school. Two participants were currently attending undergraduate courses.

Of the 22 participants, 12 had children. Ten participants had their first delivery during adolescence (from 12 to 19 years old) and all of the participants who had a childbirth experienced unintended pregnancy. Participants who had a low level of formal schooling reported that pregnancy was a relevant factor for leaving school as a 19-year-old participant who had her first child at 14 years old reported: "When I least expected it, I got pregnant. Then I had to accept it, right? I had to leave school. Now I have two children. I'm trying to look for a job here, but it's difficult because I do not have schooling." Studies reveal that adolescent pregnancy could be considered a relevant factor for discontinuity of school. In this sense, improving adolescents' SRH is essential to their social and economic well-being [17, 18]. Access to information and adolescent-friendly health services are key factors to reduce unintended early pregnancies [19–21].

Generally, women are considered to have an unmet need for contraception if they are fertile or are sexually active and want to delay or avoid becoming pregnant but are not using a contraceptive method [22]. However, in this study it is understood that an unmet need could also be expressed as lack of information about SRH and barriers to access these kind of health services. The lack of information unfolds in two main aspects: knowledge gaps about contraceptives use; and misconceptions on contraceptive methods adverse effects and their effectiveness. Communications failure and unawareness about Zika and its consequences were barriers women faced when seeking SRH in high risk areas.

The information gap on sexual and reproductive health care

Lack of knowledge on contraceptive use

Despite the increase in use of contraceptives in Brazil [23, 24], the young women were unaware of how to use them effectively – or even how to access these contraceptive methods. They reported having little trust in short acting contraceptive methods such as oral and

injectable contraceptives – which are commonly offered by the Brazilian public health services and have the highest usage prevalence in Brazil [25]. The young women justified their mistrust correlating two experiences: their personal unintended pregnancies experience and the contraceptive method used at that moment, as explained by a 16-year-old participant: "We use condoms, but we don't know the right way to use them. That is why it happens that people get pregnant when they use condoms, right? That is what happened with me."

Some young women reported that they were not using any contraceptive method, as described by an 18-yearold adolescent who experienced a repeat pregnancy within one year of the birth of her first child. Her first delivery was when she was 16 years old: "I wasn't taking anything to avoid it. I didn't know it was possible to get pregnant so fast." This finding is indicative of missed opportunities to inform adolescents about family planning, be it at school and/or during previous prenatal and postpartum care. Studies suggest that adolescents who are not actively seeking to prevent pregnancy are predisposed to it [26–28]. Population studies in Brazil indicate a large proportion of unplanned or unwanted pregnancies [5, 29–31]. Data from the National Demographic and Health Survey (PNDS) from 2016 revealed that 19% of women in Brazil experienced an unwanted pregnancy on the day of the interview; in the Northeast of the country - the region where the study was conducted - the proportion reached 22,5%. In Brazil, more than half of all births were reported as unwanted or mistimed [11, 24, 29].

The risk of an unintended pregnancy during contraceptive method changes was also not stressed and discussed with the women at risk. A 24-year-old woman who had her first delivery at 17, the second at 20 and the third delivery at 23 years old, described her experience at the family planning facility: "*They told me to use the IUD after the postpartum period. But I was only able to get out of the house after 40 days. Then I got there, and they said I could not have it inserted because more than 30 days had passed. That I would have to wait for my period to come.*" Lack of proper and up to date information about SRH, in particular, contraceptive methods use, and the risk of an unintended pregnancy leads to inadequate compliance thus adding to the dissatisfaction with family planning services.

One of the study participants did not know about the Zika virus' potential health effects or the risks of congenital syndrome. Mosquitos were described by most of them as the main vector of transmission of the Zika's illness and microcephaly in babies as one of the consequences of the virus. However, none of the young women knew about Zika's sexual transmission, or the recommendation for condom use during the pregnancy. This finding is significant for an area that was affected by the Zika epidemic and continues to be at risk for the illness.

Misconceptions on contraceptive methods adverse effects and effectiveness

Shared community perception about contraceptive methods matters in promoting its acceptance or refusal. The young women expressed concerns about the only LARC available to them in the Brazilian public health system, the copper intrauterine device (IUD). Women who fear IUD insertion justify their fright based on stories heard in their communities. Most common stories are about women who have had chronic diseases such as cancer, or women who become pregnant after the device's placement. They also explain the fear of inserting a foreign object into their bodies and about the copper decomposing inside their womb. Other misconceptions were also related as demonstrated by the literature on the supposed side effects of the IUD such as abortion, pelvic inflammatory disease, infertility or pain [32].

At the same time, the copper IUD was considered by the young women a good contraceptive choice when they heard the experiences of other women of their community. The experience of other women with whom they have interpersonal ties, close and trusting relationships is considered an important aspect before deciding whether to use the IUD. Most of those who chose the IUD sought the service after the indication of another woman who was a friend or a family member (n = 15). A 17-year-old adolescent described why she decided to use the LARC: "It was my friend who told me about the IUD and I like it. After I had my daughter, I've been thinking that I'm pregnant, all the time. So, I chose the IUD in order to make sure I'm not pregnant." Understanding the influence of the close community in young women's reproductive choices should be appreciated by the public health system and providers.

Female sterilization was understood by many young women as the only truly effective method, as described by this 16-year-old participant who became pregnant at the age of 15: "Look, I'm even afraid to have sex, I'm afraid! When I remember the amount of work this one gives me, and she is still a baby. I wanted the (tubal) ligation, but I'm too young. That's why I'm going to get the IUD and when I turn 24, I'm going to ask to get the sterilization." One possible way to understand this perception towards the irreversible contraceptive method could be due to reproductive coercion as well as stereotyping of women in health care. The category of reproductive coercion could be influencing young women's understanding of female sterilization [33, 34]. It suggests, nonetheless, that gaps in young women's knowledge about birth control have important consequences in

their fully informed choices about contraceptive methods.

The young women reported suspending the continued use of hormonal contraceptive methods due to unwanted side effects on their body or health. They also described that the misconceptions regarding the utilization of hormonal methods were related to some health care professionals who asked women to stop using pills or hormonal injections in order to continue the menstrual cycle to "cleanse the body." The young women also related that some health care providers recommended that the women stop using methods in order to switch them or simply to wait a period of time before starting a new contraceptive method. These professionals, however, did not provide any evidence-based explanations or offer an alternative method of birth control. Studies demonstrated that Brazilian women mainly among those with less years of schooling and low socioeconomic status - view menstrual bleeding as a sign of health, fertility, and of not being pregnant [35]. However, it is not yet fully understood how the women's preference for monthly menstruation can affect their sexual and reproductive health.

Barriers to access sexual and reproductive health care Challenges in overcoming communication failures

Planning when and if the young women want to experience pregnancy is an important issue, but they faced barriers to access SRH services. The wish to postpone pregnancy was highlighted by all young women who had experienced childbearing. Most of those who have a child (n = 10), sought medical advice to start a contraceptive method, only after the first childbirth. These young women, especially those with little schooling, reported experiences in which there was little understanding of the information offered by the health care professional. A 15-year-old adolescent explained the barrier she faced: "*The doctor and the nurse speak in a very complicated manner, they speak things that we cannot understand. But there is also embarrassment to ask about the same thing again.*"

Young women reported their appreciation of discussion groups and informative lectures on available contraceptive methods and SRH. They specifically identified the utilization of visual aids, such as anatomical models, and handling of samples as particularly helpful for providing information as well as for establishing rapport with health care providers. Regarding their queries about the IUD, young women affirmed that a detailed presentation about the method, with step-by-step demonstrations of its insertion in a pelvic model would be an important means for providing information and addressing their concerns. Communication barriers are one of the observed difficulties these women face when trying to access SRH in these services. Adolescent girls declared that they usually do not seek the health care facilities for family planning services. Communication failure leads to the distancing between health services and the population. One of the reasons for the high unmet contraceptive need is the inability to deliver proper information to these women and girls in a manner that is respectful and also makes sense to them.

The persistent inadequate and inefficient communication between the health care facilities and the women contributes to their perception and attitudes about the services. In addition, the community perception of a long-standing state of deficits in the public health services available lead women to view them as unreliable. The long waiting periods for appointments, lack of desired birth control methods, such as LARCs, or the communication and information gaps were reported by women as circumstances that led them to not trust that the health care services would cater to their needs. In order to overcome this barrier to access, their preemptive attitude is to first seek SRH elsewhere, such as in their close community of friends and relatives and/or with a local pharmacist.

Unawareness and misinformation on Zika

Young women reported that Zika is no longer discussed during prenatal care. Mosquito repellents are provided at the public health services to pregnant women who are beneficiaries of a governmental income transfer program for very poor families called "Bolsa familia" [6, 36]. However, the recommended use of the mosquito repellent is usually not mentioned by the health providers during the visits nor any explanation is given as to why they should use it.

Zika is also no longer considered an infection that can affect their health or the development of their newborn and fetus. Zika infection was not a concern during the pregnancy of young women who experienced childbirth in late 2017 or 2018 (n = 7). Only young women who were pregnant during the epidemic in 2015 and 2016 (n = 6) reported being worried about the potential Zika effects of vertical transmission on their fetus' health, as described by this 21-year-old participant: "*I got pregnant when the cases of babies with microcephaly were on TV. I was very afraid of Zika. Afraid that Zika would pass on to her.*" This worry ended once the infant was born with normal head circumference. Microcephaly was the only feature recognized as a consequence of Zika infection in the newborn.

These findings indicate a lack of health awareness regarding Zika virus and its effects. Zika is not perceived as a current issue that can affect women's wellbeing. Participants refute the idea that after the end of the outbreak, Zika illness would still be a threat to their pregnancy and newborns. They justified their low level of concern by the fact that Zika is no longer broadcast on radio and television, is not part of the conversations in social gatherings, and is not even brought up by health professionals.

Although the number of Congenital Zika Syndrome (CZS) cases have decreased, cases continue to be reported. In 2018, 8680 probable cases of Zika virus disease were registered in Brazil, with an incidence rate of 4.2 cases/100,000 inhabitants [37]. According to the March 2019 bulletin from the Ministry of Health, 1657 suspected cases were reported in 2018 as CZS, 227 children have been confirmed as Zika-affected cases, and 839 are still under investigation [38]. Brazilian women continue to live with the main transmitting mosquito, *Aedes aegypti*. Social and environmental conditions of these women remain unchanged and Zika illness is still present in the country, especially in the Northeastern regions.

Zika in Brazil should not only be considered in epidemiological terms, but also as a picture of social and gender inequalities which increase the vulnerability of women and families most at risk [6, 7, 39]. Women, especially adolescents, affected by Zika in Brazil represent one of the most vulnerable groups in the country; in addition, they have a late start to prenatal care and live in areas where inequities and extreme poverty persist [6]. Lack of information and barriers to access SRH services reported in this study can serve as an urgent next step for prioritized public health policies.

Methodological considerations

This study contributed to a deeper understanding of how young women's attitudes towards their SRH needs exposes the barriers to access care. Two researchers did the interview transcriptions' coding process separately and independently in order to increase trustworthiness of data analysis. In the few occasions of discrepancies during the data coding process, a third researcher who is experienced in the methodology used, analyzed the same data and discussed it with the research team in order to ensure its credibility and consistency of coding. Our study used thematic analysis to generate a conceptually informed interpretation of data thus increasing conformability and allowing for the creation of a conceptual description of women's attitudes when seeking SRH in these locations during the period studied.

Our study is limited by the design and sampling selection. Women who did not access the SRH service were not part of this study, so conclusions regarding barriers to access may be underemphasized. Generalizability is limited due to the nature of the study design and further research is needed in order to better understand the sexual and reproductive services practices as well as unmet contraceptive needs in different contexts.

Conclusions

Our data suggest that young women in Zika affected areas have knowledge gaps of Zika related illnesses and about SRH. Young women perceived that at the national level and also at the health care facility level family planning and Zika illness are no longer health concerns. The description of the young women's perceptions and challenges provides a comprehensive overview of their SRH needs and barriers to access. Lack of information on contraceptive use and its effects along with communication failures regarding SRH and unawareness about Zika create access barriers and likely contribute to the unmet SRH needs of young women living in high risk areas for Zika.

The findings presented in this paper could contribute to developing guidance on health-care services in order to enhance better provision of SRH counseling and services to young women and adolescents living in Zika affected regions. To conclude, in order to provide culturally sensitive, age appropriate and effective SRH care for young women, it is essential that these young women and girls be placed at the center of any public health response to an emergency affecting women of reproductive age.

Abbreviations

CZS: Congenital Zika Syndrome; HRP: Human Reproduction Program; IUD: Intrauterine Device; LARC: Long-acting reversible contraceptive; PHEIC: Public Health Emergency of International Concern; PNAD: Brazilian National Household Survey (in Portuguese: Pesquisa Nacional por Amostra de Domicílios); PNDS: National Demographic and Health Survey (in Portuguese: Pesquisa Nacional de Demografia e Saúde); SRH: Sexual and Reproductive Health; SSE: Sentinel Sites Evaluation; SUS: Brazilian Public Health System (in Portuguese: Sistema Único de Saúde); WHO: World Health Organization

Acknowledgments

The authors would like to acknowledge and thank all young women who gave their time in sharing their experiences on SRH issues.

Disclaimer

This report contains the collective views of an international group of experts and does not necessarily represent the decisions or the stated policy of the World Health Organization.

Authors' contributions

All who have contributed substantially to the work have been acknowledged. D Diniz, M Ali, I Ambrogi, L Brito participated equally in all stages of preparation of the manuscript. All authors have approved the submitted version of this manuscript, publication of this manuscript and agree to be accountable for the work here presented. All authors read and approved the final manuscript.

Funding

The study is funded by Large Anonymous Donor under the Grant ID 5149 for Women, Communities, and Health Systems in the Context of Zika. The funding sources did not play a role in the study design, in the writing of the report, or in the decision to submit the article for publication.

Availability of data and materials

The datasets generated during and/or analyzed during the current study are not publicly available due to ethical protection according to the International Ethical Guidelines for Health-related Research Involving Humans in order to guarantee autonomy, privacy and confidentiality regarding sensitive sexual and reproductive health matters investigated in this research, but are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

This study protocol was approved by the University of Brasilia Research Ethics Committee for Humanities and Social Sciences – CAAE: 66502017.0.0000.5540 and WHO ethics review committee. All the participants provided informed consent in order to participate in the study in accordance with Brazilian regulations which guarantees autonomy and confidentiality about their sexual and reproductive health decisions and practice.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Rights and Justice Regional Deputy Director - International Planned Parenthood Federation/Western Hemisphere Region (IPPF/WHR), Caixa Postal 8011, Brasília, DF 70094-971, Brazil. ²Department of Reproductive Health and Research (RHR), World Health Organization, Avenue Appia 20, CH-1211 Geneva 27, Switzerland. ³Researcher at Anis - Institute of Bioethics, Caixa Postal 8011, Brasília, DF 70094-971, Brazil. ⁴Director of Research at Anis -Institute of Bioethics, Caixa Postal 8011, Brasília, DF 70094-971, Brazil.

Received: 20 May 2019 Accepted: 19 January 2020 Published online: 06 February 2020

References

- Rasmussen SA, Jamieson DJ, Honein MA, Petersen LR. Zika virus and birth defects: reviewing the evidence for causality. N Engl J Med. 2016:1–7. https://doi.org/10.1056/NEJMsr1604338.
- Cauchemez S, Besnard M, Bompard P, et al. Association between Zika virus and microcephaly in French Polynesia, 2013–15: a retrospective study. Lancet. 2016:1–8. https://doi.org/10.1016/S0140-6736(16)00651-6.
- World Health Organization. Fifth meeting of the Emergency Committee under the International Health Regulations (2005) regarding microcephaly, other neurological disorders and Zika virus. https://www.who.int/en/newsroom/detail/18-11-2016-fifth-meeting-of-the-emergency-committee-underthe-international-health-regulations-(2005)-regarding-microcephaly-otherneurological-disorders-and-zika-virus. Published 2017. Accessed April 10, 2019.
- Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde. PROTOCOLO DE VIGILÂNCIA E RESPOSTA À OCORRÊNCIA DE MICROCEFALIA E/OU ALTERAÇÕES DO SISTEMA NERVOSO CENTRAL (SNC) Emergência de Saúde Pública de Importância Internacional - ESPII. Brasília; 2016. http:// combateaedes.saude.gov.br/images/sala-de-situacao/Microcefalia-Protocolode-vigilancia-e-resposta-10mar2016-18h.pdf. Accessed March 17, 2019.
- Brazil. Ministério da Saúde. SINASC Sistema de Informações de Nascidos Vivos [internet]. http://www.scielo.br/scielo.php?script=sci_arttext&pid=S01 02-311X2019001205009&lang=pt. Accessed 3 Feb 2020.
- Diniz D. Zika in Brazil: women and children at the center of the epidemic. Brasilia; 2017. http://anis.org.br/wp-content/uploads/2017/06/Zika-in-Brazil_ Anis_2017.pdf. Accessed April 10, 2019.
- Diniz D, Gumieri S, Bevilacqua BG, Cook RJ, Dickens BM. Zika virus infection in Brazil and human rights obligations. Int J Gynecol Obstet. 2017;136(1): 105–10. https://doi.org/10.1002/ijgo.12018.
- Brasil. Protocolos Da Atenção Básica : Saúde Das Mulheres. Brasilia; 2016. www.dab.saude.gov.br. Accessed April 10, 2019.
- de Leon RGP, Ewerling F, Serruya SJ, et al. Contraceptive use in Latin America and the Caribbean with a focus on long-acting reversible contraceptives: prevalence and inequalities in 23 countries. Lancet Glob Heal. 2019;7(2):e227–35. https://doi.org/10.1016/S2214-109X(18)30481-9.

Page 7 of 8

- Bahamondes L, Ali M, Monteiro I, Fernandes A. Contraceptive sales in the setting of the Zika virus epidemic. Hum Reprod. 2017;32(1):88–93. https:// doi.org/10.1093/humrep/dew310.
- Le HH, Connolly MP, Bahamondes L, Cecatti JG, Yu J, Hu HX. The burden of unintended pregnancies in Brazil: a social and public health system cost analysis. Int J Women's Health. 2014;6(1):663–70. https://doi.org/10.2147/JJWH.S61543.
- Castro MC, Han QC, Carvalho LR, Victora CG, França GVA. Implications of Zika virus and congenital Zika syndrome for the number of live births in Brazil. Proc Natl Acad Sci May. doi:https://doi.org/10.1073/pnas.1718476115.
- Marteleto LJ, Weitzman A, Coutinho RZ, Valongueiro Alves S. Women's reproductive intentions and behaviors during the Zika epidemic in Brazil. Popul Dev Rev 2017;0:1–2. doi:https://doi.org/10.1111/padr.12074.
- 14. UNFPA. Situação Da População Mundial 2017.; 2017. http://unfpa.org.br/ swop2017/swop2017.pdf.
- Corbin JM, Strauss AL, Strauss AL. Basics of qualitative research : techniques and procedures for developing grounded theory. Los Angeles: Sage Publications; 2008.
- Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3(2):77–101.
- Goodman M, Onwumere O, Milam L, Peipert JF. Reducing health disparities by removing cost, access, and knowledge barriers. Am J Obstet Gynecol. 2017;216(4):e1–382.e5. https://doi.org/10.1016/j.ajog.2016.12.015.
- Guttmacher Institute. Costs and Benefits of Meeting the Contraceptive Needs of Adolescents In Developing Regions. Vol 19. New York; 2016.
- Ehiri JE, Meremikwu A, Meremikwu MM. Interventions for preventing unintended pregnancies among adolescents. In: Ehiri JE, ed. Cochrane Database of Systematic Reviews. Chichester, UK: Wiley, 2005:126. doi:https:// doi.org/10.1002/14651858.CD005215
- 20. Liang ELM. Adolescent pregnancy : a review of the evidence. New York; 2013.
- Hadley A, Chandra-Mouli V, Ingham R. Implementing the United Kingdom Government's 10-year teenage pregnancy strategy for England (1999–2010): applicable lessons for other countries. J Adolesc Health. 2016;59(1):68–74. https://doi.org/10.1016/j.jadohealth.2016.03.023.
- Sedgh G, Ashford LS, Hussain R. Unmet Need for Contraception in Developing Countries: Examining Women's Reasons for Not Using a Method. 2016;115(24):6177–6182.
- Potter JE, Schmertmann CP, Assunção RM, Cavenaghi SM. Mapping the timing, pace, and scale of the fertility transition in Brazil. Popul Dev Rev. 2010;36(2):283–307. https://doi.org/10.1111/j.1360-0443.2009.02814.x.
- 24. Curtis SL. Contraceptive use dynamics research needs post fertility transition. Rev bras Estud Popul. 2012;29(1):191–193. doi:http://dx.doi.org/ https://doi.org/10.1590/S0102-30982012000100012.
- Farias MR, Leite SN, Tavares NUL, et al. Use of and access to oral and injectable contraceptives in Brazil. Rev Saude Publica. 2016;50(suppl 2):14s. https://doi.org/10.1590/s1518-8787.2016050006176.
- Conroy KN, Engelhart TG, Martins Y, et al. The enigma of rapid repeat pregnancy: a qualitative study of teen mothers. J Pediatr Adolesc Gynecol. 2016;29(3):312–7. https://doi.org/10.1016/jjpag.2015.12.003.
- Crosby RA, DiClemente RJ, Wingood GM, Davies SL, Harrington K. Adolescents' ambivalence about becoming pregnant predicts infrequent contraceptive use: a prospective analysis of nonpregnant African American females. Am J Obstet Gynecol. 2002;186(2):251–2. https://doi.org/10.1067/ MOB.2002.120081.
- Sheeder J, Teal SB, Crane LA, Stevens-Simon C. Adolescent childbearing ambivalence: is it the sum of its parts? J Pediatr Adolesc Gynecol. 2010; 23(2):86–92. https://doi.org/10.1016/J.JPAG.2009.06.002.
- Brasil. Ministério da Saúde. Pesquisa Nacional de Demografia e Saúde Da Criança e Da Mulher PNDS 2006 Dimensões Do Processo Reprodutivo e Da Saúde Da Criança.; 2009. http://bvsms.saude.gov.br/bvs/publicacoes/pnds_ crianca_mulher.pdf. Accessed April 3, 2019.
- Schuck-Paim C, López D, Simonsen L, Alonso W. Unintended pregnancies in Brazil - a challenge for the recommendation to delay pregnancy due to Zika. PLoS Curr 2016. doi:10.1371/currents.outbreaks. 7038a6813f734c1db547240c2a0ba291.
- Theme-Filha MM, Baldisserotto ML, Fraga ACSA, Ayers S, da Gama SGN, Leal MD. Factors associated with unintended pregnancy in Brazil: cross-sectional results from the Birth in Brazil National Survey, 2011/2012. Reprod Health. 2016;13(S3):118. https://doi.org/10.1186/s12978-016-0227-8.
- Russo JA, Miller E, Gold MA. Myths and misconceptions about long-acting reversible contraception (LARC). J Adolesc Health. 2013;52(4):S14–21. https:// doi.org/10.1016/JJADOHEALTH.2013.02.003.

- O'Connell C, Zampas C. The human rights impact of gender stereotyping in the context of reproductive health care. Int J Gynecol Obstet. 2019;144(1): 116–21. https://doi.org/10.1002/ijgo.12693.
- Grace KT, Anderson JC. Reproductive coercion: a systematic review. Trauma Violence Abuse. 2018;19(4):371–90. https://doi.org/10.1177/ 1524838016663935.
- Makuch MY, Osis MJD, Petta CA, de Pádua KS, Bahamondes L. Menstrual bleeding: perspective of Brazilian women. Contraception. 2011;84(6):622–7. https://doi.org/10.1016/j.contraception.2011.03.010.
- Brazil. Lei n. 10836, de 9 de janeiro de 2004. Cria o Bolsa Família e dá outras providências 2004. http://www.planalto.gov.br/ccivil_03/_Ato2004-2006/ 2004/Lei/L10.836.htm. Accessed April 4, 2019.
- 37. Brazil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Boletim Epidemiológico N. 04. Monitoramento Dos Casos de Dengue, Febre de Chikungunya e Doença Aguda Pelo Vírus Zika Até a Semana Epidemiológica 52 de 2018. Vol 50. Brasília2019 http:// portalarquivos2saudegovbr/images/pdf/2019/janeiro/28/2019-002pdf Accessed April 10, 2019.
- 38. Brazil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Boletim Epidemiológico N. 08. Monitoramento integrado de alterações no crescimento e desenvolvimento relacionadas à infecção pelo vírus Zika e outras etiologias infecciosas, até a Semana Epidemiológica 52 de 2018. Vol 50. Brasília; 2019. http://portalarquivos2.saude.gov.br/images/pdf/2019/ marco/22/2019-001.pdf. Accessed April 10, 2019.
- 39. Diniz D, Ambrogi I. Research ethics and the Zika legacy in Brazil. Dev World Bioeth. 2017;17(3):142–3. https://doi.org/10.1111/dewb.12175.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

